Author's response to reviews

Title: The contribution of childhood circumstances, current circumstances and health behaviour to educational health differences in early adulthood

Authors:

Laura Kestilä (laura.kestila@thl.fi)
Tuija Martelin (tuija.martelin@thl.fi)
Ossi Rahkonen (ossi.rahkonen@helsinki.fi)
Tommi Härkänen (tommi.harkanen@thl.fi)
Seppo Koskinen (seppo.koskinen@thl.fi)

Version: 6 Date: 18 May 2009

Author's response to reviews:

Dear Editors,

We have now revised the manuscript according to your requests.

Sincerely yours,
Laura Kestilä